Author's response to reviews

Title: Duration of unemployment and depression: a cross-sectional survey in Lithuania

Authors:

  Mindaugas Stankunas (m.stankunas@one.lt)
  Ramune Kalediene (socmed@kmu.lt)
  Skirmante Starkuviene (smante@centras.lt)
  Violeta Kapustinskiene (socmed@kmu.lt)

Version: 3 Date: 30 June 2006

Author's response to reviews: see over
Cover letter on the article

“Duration of unemployment and depression: a cross-sectional survey in Lithuania”

To the Editor

In the methods section of the paper we added the sentence regarding the approval from our university ethics center. The manuscript was revised, and we hope that it fits the requirements of the journal.

We would like to thank and to express our gratitude to the Editor and reviewers for their insights and comments, what contributed to the improvement of the paper. We hope that the paper will be interesting not only for the researchers in this field, but also will attract a wider audience.

Sincerely,
On behalf of authors,
Mindaugas Stankunas